Author’s response to reviews

Title: Vitamin D in children with growth hormone deficiency due to pituitary stalk interruption syndrome

Authors:

CECILE DELECROIX (ceciledelecroix88@gmail.com)
RAJA BRAUNER (raja.brauner@wanadoo.fr)
JEAN-CLAUDE SOUBERBIELLE (jean-claude.souberbielle@aphp.fr)

Version: 2 Date: 30 Dec 2017

Author’s response to reviews:

Dear Pr Iughetti,

Thank you for having reviewed our manuscript and for your message.

We included the cut-offs defining the hypothalamic-pituitary deficiencies and the reference 2.

The paragraph “Declarations” is presented according to the recommendations of BMC Pediatrics.

The English language has been edited twice by the AJE (certificates joined).

Best wishes.

Pr Raja Brauner